Author's response to reviews


Authors:

PILAR CARRASCO-GARRIDO (pilar.carrasco@urjc.es)
ANA LOPEZ DE ANDRÉS (ana.lopez@urjc.es)
VALENTÍN HERNANDEZ-BARRERA (valentin.hernandez@urjc.es)
ISABEL JIMÉNEZ-TRUJILLO (isabel.jimenez@urjc.es)
CÉSAR FERNANDEZ-DE-LAS-PEÑAS (cesar.fernandez@urjc.es)
DOMINGO PALACIOS-CEÑA (domingo.palacios@urjc.es)
SOLEDAD GARCÍA-GÓMEZ-HERAS (soledad.garcia@urjc.es)
RODRIGO JIMÉNEZ-GARCÍA (rodrigo.jimenez@urjc.es)

Version: 2 Date: 10 April 2014

Author's response to reviews:

Madrid, 10 April 2014
Editorial Office
BMC Pharmacology and Toxicology

Dear Editor:

I enclose the manuscript, “PREDICTIVE FACTORS OF SELF-MEDICATED ANALGESIC USE IN SPANISH ADULTS: A POPULATION-BASED NATIONAL CROSS-SECTIONAL STUDY.” with the following changes:

(1) Title Page: The email addresses of all authors have been included.
(2) Line Numbering: The lines and page numbers have been included.

Thanking you in advance for your attention in this matter,

Yours faithfully,

Profa. Pilar Carrasco Garrido (PhD, MPH)

E-mail: pilar.carrasco@urjc.es